\begin{table} 
\caption{\label{tab9} SN~Ia template spectra in ${\cal SN}$-Fit Database.} 
%\centerline
{\begin{tabular}{cccc}
\hline\hline 
SN & Type & Number of & Reference\\ 
Name && spectra &\\
\hline
\object{SN~1981B} & Ia       & 6  &\citet{Branch83}\\ 
\object{SN~1986G} & 91bg-like  & 5  &\citet{Phillips87}\\
\object{SN~1990N} & Ia       & 7  &\citet{Leibundgut91,Phillips92,Mazzali93}\\
\object{SN~1991T} & 91T-like & 9  &\citet{Filippenko92}\\
\object{SN~1991bg} & 91bg-like & 6  &\citet{Phillips92,Mazzali95}\\
\object{SN~1992A} & Ia       & 12 &\citet{Kirshner93}\\
\object{SN~1994D} & Ia       & 18 &\citet{Meikle96,Patat96}\\
\object{SN~1996X} & Ia       & 6  &\citet{Salvo01}\\  
\object{SN~1997br} & 91T-like & 10 &\citet{Li99}\\     
\object{SN~1997cn} & 91bg-like & 1  &\citet{turatto98}\\
\object{SN~1998bu} & Ia      & 12 &\citet{Jha99}\\    
\object{SN~1999aa} & 91T-like & 10 &\citet{Garavini04}\\
\object{SN~1999by} & 91bg-like & 14 &\citet{Garnavich04}\\
\object{SN~1999ee} & Ia      & 11 &\citet{Hamuy02}\\  
\object{SN~2000cx} & 91T-like & 24 &\citet{Li01c}\\    
\object{SN~2002bo} & Ia      & 12 &\citet{Benetti04}\\
\object{SN~2003du} & Ia      & 4  &\citet{Anupama05}\\
\hline
Total peculiar Ia = 79&   Total "normal Ia'' = 88&  Total = 167.\\
\end{tabular}}
\par
\smallskip
\par
\end{table}